Author's response to reviews

Title: Physical activity and renal cell carcinoma among black and white Americans: a case-control study

Authors:

Qian Xiao (xiaoq2@mail.nih.gov)
Linda Liao (donglm@mail.nih.gov)
Charles E Matthews (matthewsce@mail.nih.gov)
Wong-Ho Chow (wchow@mdanderson.org)
Faith G Davis (fdavis@ualberta.ca)
Kendra Schwartz (kensch@med.wayne.edu)
Mark P Purdue (purduem@mail.nih.gov)
Jonathan N Hofmann (hofmannjn@mail.nih.gov)
Joanne S Colt (coltj@exchange.nih.gov)

Version: 4
Date: 4 August 2014

Author's response to reviews: see over
Dear editor:

Thank you for the opportunity for us to further revise our manuscript, entitled “Physical activity and renal cell carcinoma among black and white Americans: a case-control study” for publication in *BMC Cancer*. We have included the name of all the institutions that approved our study in the methods section (page: “Study procedures were approved by Institutional Review Boards at National Cancer Institute, University of Illinois at Chicago, Wayne State University, and Westat, Inc.”) We have updated the author contribution section and moved it before the acknowledgements section. Finally, we have changed “introduction” into “background”.

Sincerely,

Qian Xiao

Division of Cancer Epidemiology and Genetics
National Cancer Institute
9609 Medical Center Dr
Rockville, Maryland 20850